Author’s response to reviews

Title: Alarming prevalence and clustering of modifiable noncommunicable disease risk factors among adults in Bhutan: a nationwide cross-sectional community survey

Authors:

Dorji Pelzom (dorji.pelzom@gmail.com)
Petros Isaakidis (Petros.Isaakidis@joburg.msf.org)
Myo Oo (dr.myominnoo@gmail.com)
Mongal Gurung (msgurung@health.gov.bt)
Pemba Yangchen (pembayangchen7@gmail.com)

Version: 4 Date: 07 Dec 2017

Author’s response to reviews:

Dear Natalie Pafitis,

Thank you again for your valuable feedbacks on the manuscript. I have done the revision as suggested as follows:

Editor Comments:

(1) Please copy edit your 'Authors' Contributions' for improved quality of written English. Please ensure to correct the statement 'The have also been involved in drafting the manuscript....' to 'They have also been involved in drafting the manuscript...'. Where you refer to two authors (eg: PI MMO), please separate the initials with a comma (eg: PI, MMO).

Thank you for the comment, I have changed the authors’ contributions section as per the suggestion at page 18 lines 389, 390 and 392

(2) Re-write you Acknowledgements section so that it is clear who you are thanking and why.

Thank you for this suggestion, I have re-wrote the acknowledgement section at page 19 lines 397-409.
Thank you again, looking forward to hear from you soon.

Warm wishes

Dorji Pelzom